Author’s response to reviews


Authors:

Paul Arias Medina (paul.arias@ucuenca.edu.ec)

Version: 6 Date: 30 Jun 2019

Author’s response to reviews:

Dear editor,

I have updated the Ethics section:

The Problems, Expectations and Aspirations of Children (PEACH) Survey was approved by the International Migration and Local Development Project of the University of Cuenca and the Institute of Development Policy (IOB) of the University of Antwerp. IOB Bureau Members: Tom De Herdt, Nathalie Holvoet, Filip Reyntjens, Sara Greenen, and Hugo De Craen.

The data collection process complied with Ecuadorian national guidelines. A cooperation agreement was signed between the Ministry of Education and the University of Cuenca.

Parents/legal guardians of children agreed to participate by signing a letter prior to the data collection.